IE 2030 Lecture 3 slide 5 ERGONOMICS

Interaction of human body with machines. Lifting, carrying, turning

Heights of belts, sizes of packages at UPS. finger and armspan, knob size.

CTRL-ALT-DEL.

Ergonomics ! Human-Machine Systems, Computer Interfaces

Number of windows open at once Mouse, clicking, dragging, pulldown windows Defaults Help

IE 2030 Lecture 3 slide 6 Dark Side of I.E.

Taylor, the rest of the story.

False claims. Reputation built on high speed steel.

Frederick Taylor sleeps on his back. Meat machine vs. cognitive being concepts Why I.E.s are hated.

Scene, p. 98 of Models text.

People don’t like to be told what to do.

And often they are right, too.